Author's response to reviews

Title: Resection of a malignant paraganglioma located behind the retrohepatic segment of the inferior vena cava

Authors:

Changjun Jia (jiacj sj@163.com)
Xinlu Wang (wangxl1@sj-hospital.org)
Chaoliu Dai (daicl sj@163.com)
Xianmin Bu (buxianmin@163.com)
Songlin Peng (ps1514@126.com)
Feng Xu (xufeng sy@yahoo.com.cn)
Yongqing Xu (xuyq1@sj-hospital.org)
Yang Zhao (zy1@sj-hospital.org)

Version: 2  Date: 11 January 2013

Author's response to reviews:

Editorial Office,
BMC Surgery
01, 11, 2013

Dear Editor,

We would like to submit a manuscript entitled “Resection of a malignant paraganglioma located behind the retrohepatic segment of the inferior vena cava” from Chaoliu Dai etc. for consideration for publication in BMC Surgery.

This paper has not been published elsewhere in whole or in part. All authors have read and approved the content, and agree to submit it for consideration for publication in your journal. There are no ethical/legal conflicts involved in the article. This manuscript has been edited and proofread by Medjaden Bioscience Limited.

Your consideration for this manuscript is highly appreciated.
Looking forward to hearing from you soon.

Yours sincerely,

Dr. Chaoliu Dai
Institution: Department of General Surgery, Shengjing Hospital of China Medical University, Shenyang 110004, China
Add: No.36 Sanhao Street, Heping District, Shenyang 110004, China
Tel: +86-24-96615-31511